Author's response to reviews

Title: Eliciting parental support for the use of newborn blood spots for pediatric research

Authors:

Edwina Yeung (yeungedw@mail.nih.gov)
Germaine Buck Louis (louisg@mail.nih.gov)
David Lawrence (david.lawrence@health.ny.gov)
Kurunthachalam Kannan (krunthachalam.kannan@health.ny.gov)
Alexander McLain (mclaina@mailbox.sc.edu)
Michele Caggana (michele.caggana@health.ny.gov)
Charlotte Druschel (charlotte.druschel@health.ny.gov)
Erin Bell (ebell@albany.edu)

Version: 6 Date: 27 January 2016

Author's response to reviews: see over
January 25, 2015

Giulia Mangiameli
Executive Editor
BMC Medical Research Methodology

Dear Editor Mangiameli:

Enclosed is our revised manuscript titled “Eliciting parental support for the use of newborn blood spots for pediatric research” for consideration for publication as a research article in BMC Medical Research Methodology. We have now acknowledged that “Findings were presented in abstract and poster form to the American College of Epidemiology Annual meeting, Silver Spring, MD, Sept 8, 2014.”

Sincerely,

Edwina Yeung, PhD, ScM